Author’s response to reviews

Title: Educational Potential of a Virtual Patient System for Caring for Traumatized Patients in Primary Care

Authors:

Solvig Ekblad (Solvig.Ekblad@ki.se)
Richard F Mollica (RMOLLICA@PARTNERS.ORG)
Uno Fors (uno@dsv.su.se)
Ioannis Pantziaras (Ioannis.Pantziaras@ki.se)
James Lavelle (JLAVELLE@PARTNERS.ORG)

Version: 3 Date: 27 June 2013

Author’s response to reviews: see over
Dear Editor of BMC Medical Education,

First we would like to take the opportunity to warmly congratulate you to your very interesting journal BMC Medical Education. It covers a broad area in the field of medical education. We are very pleased to realize that one of your interest areas encompasses simulation-based medical education – a field that we share. We hope therefore to contribute further to this area of medical education, by sending our revised version MS: 9168007469104745 “Educational Potential of a Virtual Patient System for Caring for Traumatized Patients in Primary Care” for your final consideration. Authors are: Solvig Ekblad, Richard F Mollica, Uno Fors, Ioannis Pantziaras and James Lavelle. To our knowledge, this is the first study examining these attributes in regard to a dedicated refugee trauma VP.

We would like to ensure you that this manuscript represents results from original work. The submitted material has not been published and it is not being considered for publication elsewhere. All authors have contributed significantly, have critically reviewed its content and have approved the final revised version submitted. Further, there are no conflicts of interest related to the research reported in this manuscript.

We are looking forward to your response and truthfully hope that this manuscript meets the standards of BMC Medical Education.

Yours sincerely,

Solvig Ekblad, PhD, Associate professor
Head of Cultural Medicine Unit
Corresponding author

Address to corresponding author:
Department LIME, Cultural Medicine Unit
Karolinska Institutet
Tomtebodavägen 18A, plan 3
SE-171 77 Stockholm, Sweden
E-mail: Solvig.Ekblad@ki.se